**Author's response to reviews**

**Title:** Brucella bacteremia in patients with acute leukemia: two case reports

**Authors:**

Khalid A Al-Anazi (khalid_alanazi@yahoo.com)
Asma M Al-Jasser (asjass2002@yahoo.com)

**Version:** 2 **Date:** 5 August 2007

**Author's response to reviews:**

Prof. Michael Kid
Editor-in-Chief
Journal of Medical Case Reports

Dear Sir:

We are sending you the revised version of our manuscript [Brucella bacteremia in patients with acute leukemia: two case reports] for consideration of publication in your distinguished international medical journal.

All the changes and modifications requested were induced:

1- The title of the manuscript was modified so as to end up in case report.
2- The sub-title case reports was changed to case presentations.
3- The number of references was reduced to 15 and consequently few modifications were introduced in the background and discussion section.
4- The typing and spelling mistakes were corrected.
5- The requested/suggested modification to the consent section was made.

Thank you very much for your valuable suggestions

Best regards and wishes

Khalid Al-Anazi and Asma Al-Jasser